Author's response to reviews

Title: Higher self-reported prevalence of hypertension among Moluccan-Dutch than among the general population of the Netherlands: results from a cross-sectional survey

Authors:

Junus M van der Wal (junusvanderwal@live.nl)
Adee J Bodewes (a.j.bodewes@amc.uva.nl)
Charles O Angyemang (c.o.angyemang@amc.uva.nl)
Anton E Kunst (a.kunst@amc.uva.nl)

Version: 2
Date: 10 September 2014

Author's response to reviews: see over
Cover letter

Re-submission to: BMC Public Health

Title of Paper: “Higher self-reported prevalence of hypertension among Moluccan-Dutch than among the general population of the Netherlands: results from a cross-sectional survey”

Corresponding author: Junus van der Wal

Author address: Department of Public Health, Academic Medical Centre, University of Amsterdam, Meibergdreef 9, 1105 AZ Amsterdam, The Netherlands

Author email address: junusvanderwal@live.nl

On behalf of my co-authors and myself, I re-submit the enclosed manuscript for consideration by BMC Public Health. I have revised the manuscript as requested by the editorial office of the journal. The following revision were made:

- The main text of this article now contains line numbers (page numbers were already present).
- The manuscript now contains a statement regarding possible conflicts of interests of the authors of this article.
- The method section of the article now contains a statement regarding ethical approval.

This manuscript has not been published in this or a substantially similar form (in print or electronically, including on a web site), nor accepted for publication elsewhere, nor is it under consideration by another publication.

Kind regards,

Junus van der Wal